Author’s response to reviews

Title: Combined mutations of asxl1, cbl, flt3, idh1, idh2, jak2, kras, npm1, nras, runx1, tet2 and wt1 genes in myelodysplastic syndromes and acute myeloid leukemias

Authors:

Julien Rocquain (rocquainju@marseille.fnclcc.fr)
Nadine Carbuccia (carbuccian@marseille.fnclcc.fr)
Virginie Trouplin (trouplinv@marseille.fnclcc.fr)
Stephane Raynaud (raunauds@marseille.fnclcc.fr)
Anne Murati (muratia@marseille.fnclcc.fr)
Meyer Nezri (nezri.mrdint@ch-martigues.fr)
Zoulika Tadrist (zoulika.tadrist@ch-salon.fr)
Sylviane Olschwang (olschwangs@marseille.fnclcc.fr)
Norbert Vey (veyn@marseille.fnclcc.fr)
Daniel Birbaum (daniel.birbaum@inserm.fr)
Veronique Gelsi-Boyer (gelsiv@marseille.fnclcc.fr)
Marie-Joelle Mozziconacci (mozziconaccimj@marseille.fnclcc.fr)

Version: 3 Date: 15 July 2010

Author’s response to reviews:

Dear Editor,

We agree with the reviewer. To make sure his two points are strongly made in the manuscript we included them at three different places: in the abstract, in the discussion and in the conclusion. At each of these places we mentioned that the model is speculative and that it cannot be applied outside what was tested, i.e. AMLs without balanced translocation and complex karyotype. These modifications are underlined.

Thank you very much for allowing us to improve the manuscript.

Sincerely,

Dr. Marie-Joelle Mozziconacci